Author’s response to reviews

Title: Absence of herb-drug interactions of Mistletoe with the Tamoxifen metabolite (E/Z)-Endoxifen and Cytochrome P450 3A4/5 and 2D6 in vitro

Authors:

Ulrike Weissenstein (u.weissenstein@vfk.ch)
Matthias Kunz (Matthias.Kunz@iscador.ch)
Mouhssin Oufir (mouhssin.oufir@unibas.ch)
Jue Theresa Wang (juetheresa.wang@unibas.ch)
Matthias Hamburger (Matthias.Hamburger@unibas.ch)
Konrad Urech (k.urech@vfk.ch)
Ursula Regueiro (Ursula.Regueiro@iscador.ch)
Stephan Baumgartner (s.baumgartner@vfk.ch)

Version: 3 Date: 09 Jan 2019

Author’s response to reviews:

Dear Emily Wollmuth

We were pleased that our manuscript “Absence of herb-drug interactions of mistletoe with the Tamoxifen metabolite (E/Z)-Endoxifen and Cytochrome P450 3A4/5 and 2D6 in vitro” (manuscript BCAM-D-18-00525R1) is potentially acceptable.

We carried out following requested revisions:

- Reformulation of text overlaps
- Inclusion of generated and analyzed datasets as supplementary files
- Adaptation of availability of data
- Inclusion of Acknowledgements
- Upload of missing additional files
Upload of the clean manuscript

Thank you for the positive and helpful suggestions.

Sincerely,

Ulrike Weissenstein, PhD
(Corresponding author)